Author's response to reviews

Title: Feasibility of three wearable sensors for 24 hour monitoring in middle-aged women

Authors:

Jennifer L Huberty (jhuberty@asu.edu)
Diane K Ehlers (dkehlers@asu.edu)
Jonathan Kurka (jkurka@asu.edu)
Barbara Ainsworth (barbara.ainsworth@asu.edu)
Matthew P Buman (matthew.Buman@asu.edu)

Version: 2 Date: 1 August 2014

Author's response to reviews: see over
August 1, 2014

To Whom It May Concern:

Please accept the manuscript entitled, “Feasibility of three behavioral monitors for 24-hour monitoring of sleep, sedentary, and active behaviors in middle-aged women” to BMC Women’s Health. **The authors believe that this paper is innovative and adds to the literature** because it is, to our knowledge, the first paper to assess feasibility of three behavioral monitors in a sample of middle-aged women. The paper provides important information that can be used to inform behavioral interventions as well as the design of future behavioral monitors in middle-aged women.

All authors have read and approved the final paper. There are no conflicts of interest. The work was supported and approved by the Institutional Review Board at Arizona State University.

We look forward to the opportunity to have our work published in BMC Women’s Health.

Sincerely,

[Signature]

Jennifer Huberty, PhD